Author's response to reviews

Title: Mirroring everyday clinical practice in clinical trial design: a new concept to improve the external validity of randomized double-blind placebo-controlled trials in the pharmacological treatment of major depression

Authors:

Emanuel Severus (Emanuel.Severus@uniklinikum-dresden.de)
Florian Seemüller (Florian.Seemueller@med.uni-muenchen.de)
Michael Berger (M.Berger@med.uni-muenchen.de)
Sandra Dittmann (Sandra.Dittmann@med.uni-muenchen.de)
Michael Obermeier (Michael.Obermeier@med.uni-muenchen.de)
Andrea Pfennig (Andrea.Pfennig@uniklinikum-dresden.de)
Michael Riedel (Riedel@med.uni-muenchen.de)
Sophia Frangou (sophia.frangou@kcl.ac.uk)
Hans-Juergen Möller (Hans-Juergen.Moeller@med.uni-muenchen.de)
Michael Bauer (Michael.Bauer@uniklinikum-dresden.de)

Version: 3 Date: 6 June 2012

Author's response to reviews: see over
6th of June, 2012

Dear Dr. D'Souza,

Attached please find our revised manuscript “Mirroring everyday clinical practice in clinical trial design: a new concept to improve the external validity of randomized double-blind placebo-controlled trials in the pharmacological treatment of major depression” which we formatted according to instructions for authors. Furthermore we ensured that it conforms to the journal style of BMC Medicine.

Sincerely yours,

W. Emanuel Severus, M.D.